Author’s response to reviews

Title: Adenovirus type 7 associated with severe and fatal acute lower respiratory infections in argentinian children

Authors:

Guadalupe Carballal (gcarballal@cemic.edu.ar)
Cristina Videla (cvidela@cemic.edu.ar)
Alicia Misirlian (aliciamisir@arnet.com.ar)
Paula Requeijo (prequeijo@hotmail.com)
Maria del Carmen Aguilar (mcaguilar@hotmail.com)

Version: 4 Date: 5 Jul 2002

According with to the editorial suggestion the following changes had been made.
First page: Dr Carballals institutional affiliations to the author with the use of superscript numbers was linKed.
Abstract: was shortened and subsectioned into Background, Methods, Results and Conclusions
Competing Interests was included.
Author Contributionship was included.
Figure legends was included.
Figures were modified and cropped.